Author's response to reviews

Title: Prevalence and risk factors of premature coronary artery disease in patients undergoing coronary angiography in Kurdistan, Iraq

Authors:

Ameen Mosa Mohammad Dr (doctoramb@yahoo.com)
Hekmat Izzat Jehangeer Dr (Hekmatizzat@yahoo.com)
Sabri Khalif Sheikhow Dr (doctorammb@gmail.com)

Version: 4 Date: 30 October 2015

Author's response to reviews: see over
The author's response to reviews

Title: Prevalence and Risk Factors of Premature Coronary Artery Disease in Patients Undergoing Coronary Angiography in Kurdistan, Iraq

October 9th, 2015

To editors,

BMC Cardiovascular Disorders. (Coronary section)

Editor's comments:

1) Post-review copyediting

Author response: The manuscript was fully proofread again.

2) STROBE Guidelines

Author response: Compared with the checklist of STROBE, the manuscript is checked and made fully accorded.

3) Acknowledgements

Author response: The paper is completely made available by the authors themselves. No institution funded the publication of the paper. The BMC was kind enough to waive the cost of publication.

4) Please move the version of fig 1…

Author response: Done

5) Please change the title of the Disclosures section to “Competing Interests”

Author response: Done

Many thanks,

Ameen Mosa Mohammad,
Professor (assistant) of Cardiology
MBCHB, FICMS (med.), F.I.C (cardio),
Department of Medicine, Medical School
Faculty of Medical Sciences, University of Duhok, Kurdistan, Iraq
Corresponding Author (E-mail: doctoramb@yahoo.com)